Author's response to reviews

Title: Stages of change: A qualitative study on the implementation of a perinatal audit program in South Africa

Authors:

Maria Belizán (maria.belizan@gmail.com)
Anne-Marie Bergh (anne-marie.bergh@up.ac.za)
Carolé Cilliers (22573143@nwu.ac.za)
Robert C Pattinson (robert.pattinson@up.ac.za)
Anna Voce (voceas@ukzn.ac.za)

Version: 12 Date: 25 September 2011

Author's response to reviews:

Dear Editors

I am uploading the final version of the manuscript with the changes required by you. I have two comments:

• We changed the heading “Findings” to “Results”, although it is preferred that qualitative research studies report on findings rather than results.
• You refer to merged cells in our table, but there are not such cells. The vertical lines in the table have however been removed.

Yours sincerely
Anne-Marie Bergh